Author's response to reviews

Title: Preventative tele-health supported services for early stage Chronic Obstructive Pulmonary Disease: a pragmatic randomized controlled trial pilot

Authors:

Deborah Fitzsimmons (d.a.fitzsimmons@sheffield.ac.uk)
Jill Thompson (jill.thompson@sheffield.ac.uk)
Mark Hawley (mark.hawley@sheffield.ac.uk)
Gail Mountain (g.a.mountain@sheffield.ac.uk)

Version: 6 Date: 5 January 2011

Author's response to reviews:

Changes made to second author name and to correct reference numbering as requested.